November 1st, 2011

The Journal of Medical Case Reports Editorial Team

Manuscript number: 1333955775694612  
Corresponding author: Racette  
Title: Pregnancy in multiple system atrophy: a case report

Dear Editorial Team,

We have obtained written informed consent from the patient’s next of kin (her husband) for publication of the case. The consent statement of the manuscript has been revised accordingly.

Sincerely,

Lirong Zhu, M.D.,Ph.D.  
Nigel J. Cairns, Ph.D.  
Samer D. Tabbal, M.D.  
Brad A. Racette, M.D.